Treatment guided by fractional exhaled nitric oxide in addition to standard care in 6- to 15-year-olds with asthma: the RAACENO RCT

Steve Turner,¹ Seonaidh Cotton,² Jessica Wood,² Victoria Bell,² Edwin-Amalraj Raja,³ Neil W Scott,³ Heather Morgan,⁴ Louisa Lawrie,² David Emele,² Charlotte Kennedy,⁵ Graham Scotland,⁵ Shona Fielding,⁶ Graeme MacLennan,² John Norrie,⁷ Mark Forrest,² Erol Gaillard,⁸ Johan de Jongeste,⁹ Marielle Pijnenburg,⁹ Mike Thomas¹⁰ and David Price^{11,12}

¹Royal Aberdeen Children's Hospital, University of Aberdeen, Aberdeen, UK
²Health Services Research Unit, University of Aberdeen, Aberdeen, UK
³Medical Statistics Team, University of Aberdeen, Aberdeen, UK
⁴Institute of Applied Health Sciences, University of Aberdeen, Aberdeen, UK
⁵Health Economics Research Unit, University of Aberdeen, Aberdeen, UK
⁶Medical Statistics Team, University of Aberdeen, Aberdeen, UK
⁶Medical Statistics Team, University of Aberdeen, Aberdeen, UK
⁸Leicester Nilke Biomedical Research Centre, University of Leicester, Leicester, UK
⁹Erasmus University Medical Centre, Rotterdam, the Netherlands
¹⁰Primary Care and Population Sciences, University of Southampton, Southampton, UK
¹¹Observational and Pragmatic Research Institute, Singapore
¹²Centre of Academic Primary Care, University of Aberdeen, Aberdeen, UK

*Corresponding author s.w.turner@abdn.ac.uk

Declared competing interests of authors: Steve Turner reports that Circassia (Oxford, UK) lent 16 NIOX VERO® devices to measure nitric oxide in study participants and provided, at no cost, associated consumables for these devices. Seonaidh Cotton reports grants from the National Institute for Health and Care Research (NIHR) Health Technology Assessment (HTA) programme outside the submitted work. Graham Scotland reports non-financial support from Merck KgaA (Darmstadt, Germany) outside the submitted work. Graeme MacLennan reports grants from the NIHR HTA programme outside the submitted work. John Norrie reports grants from the University of Aberdeen and the University of Edinburgh during the conduct of the study and declares membership of the following NIHR boards: CPR Decision-making Committee (2016), HTA Commissioning Board (2010–16), HTA Commissioning Sub-Board (EOI) (2012–16), HTA Funding Boards Policy Group (2016), HTA General Board (2016–19), HTA Post-Board funding teleconference (2016–19), NIHR Clinical Trials Unit Standing Advisory Committee (2017–present), NIHR HTA and Efficacy and Mechanism Evaluation (EME) Editorial Board (2014–19), and the Pre-exposure Prophylaxis Impact Review Panel (2017–present). Erol Gaillard reports consultancy work for Boehringer Ingelheim (Ingelheim am Rhein, Germany), with money paid

to the institution (University of Leicester); an investigator-led research grant from Circassia, Gilead (Foster City, CA, USA) and Chiesi Ltd (Manchester, UK); and research collaboration with Medimmune (Gaithersburg, MD, USA). David Price has board membership with AstraZeneca (Cambridge, UK), Boehringer Ingelheim, Chiesi, Mylan (Canonsburg, PA, USA), Novartis (Basel, Switzerland), Regeneron Pharmaceuticals (Tarrytown, NY, USA), Sanofi Genzyme (Cambridge, MA, USA), Thermofisher (Waltham, MA, USA) and Zentiva (Sanofi Generics) (Prague, Czech Republic); holds consultancy agreements with Airway Vista Secretariat (Seoul, South Korea), AstraZeneca, Boehringer Ingelheim, Chiesi, EPG Communication Holdings Ltd (Kent, UK), FIECON Ltd (St Albans, UK), Fieldwork International (London, UK), GlaxoSmithKline (Brentford, UK), Mylan, Mundipharma (Limburg an der Lahn, Germany), Novartis, OM Pharma SA (Meyrin, Switzerland), PeerVoice (London, UK), Phadia AB (Uppsala, Sweden), Spirosure Inc. (Pleasanton, CA, USA), Strategic North Ltd (Hale, UK), Synapse Research Management Partners (Barcelona, Spain), Talos Health Solutions (Brussels, Belgium), Theravance (South San Francisco, CA, USA) and WebMD Global LLC (New York, NY, USA); has received grants and unrestricted funding for investigator-initiated studies [conducted through Observational and Pragmatic Research Institute Pte Ltd (Norwich, UK)] from AKL Research and Development Ltd (Stevenage, UK), AstraZeneca, Boehringer Ingelheim, British Lung Foundation (London, UK), Chiesi, Mylan, Novartis, Regeneron Pharmaceuticals, Respiratory Effectiveness Group (Ely, UK), Sanofi Genzyme, Theravance, UK NHS and Zentiva (Sanofi Generics); has received payment for lectures/speaking engagements from AstraZeneca, Boehringer Ingelheim, Chiesi, Cipla (Mumbai, India), GlaxoSmithKline, Kyorin, Merck (Kenilworth, NJ, USA), Mylan, Mundipharma, Novartis, Regeneron Pharmaceuticals, Sanofi Genzyme and Zentiva (Sanofi Generics); has received payment for travel/accommodation/meeting expenses from AstraZeneca, Boehringer Ingelheim, Mundipharma, Mylan, Novartis, Thermofisher; has received stock/stock options from AKL Research and Development Ltd, which produces phytopharmaceuticals; owns 74% of the social enterprise Optimum Patient Care Ltd (Australia and UK) and 92.61% of Observational and Pragmatic Research Institute Pte Ltd (Singapore); has a 5% shareholding in Timestamp (Tel Aviv-Yafo, Israel), which develops adherence monitoring technology; is peer reviewer for the grant committees of the UK EME and HTA programmes; and was an expert witness for GlaxoSmithKline.

Published May 2022 DOI: 10.3310/AWOI5587

Plain English summary

The RAACENO RCT Efficacy and Mechanism Evaluation 2022; Vol. 9: No. 4 DOI: 10.3310/AWOI5587

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

We wanted to find out whether or not a simple breath test (to measure how much nitric oxide is breathed out) might help doctors to decide the best treatment for children with asthma. Better treatment might reduce the chance of children having an asthma attack.

In total, 509 children aged 6–15 years took part in the study. They either had their asthma treatment guided by their symptoms in the usual way or had their asthma treatment guided by their asthma symptoms plus their nitric oxide result. Information about symptoms and nitric oxide was entered into a computer system that made a recommendation about treatment (to keep treatment the same, or to increase or decrease the treatment). The doctors could follow this recommendation or make a different decision about treatment. Children were in the study for 12 months and had a treatment recommendation when they joined the study and then every 3 months. While they were in the study, we collected information about any asthma attacks that they had. The average age of the children in the study was 10 years and 60% were boys.

There was no difference in the proportion of children who had an asthma attack when comparing those who had their asthma treatment guided by symptoms alone with those who had their asthma treatment guided by symptoms plus their nitric oxide result. We looked at several other asthma outcomes (e.g. asthma symptoms, emergency visits and breathing tests) and none of these was different between treatment groups.

This means that adding the breath test to measure nitric oxide did not help doctors to decide the best treatment for children with asthma.

We interviewed some families and staff involved in the RAACENO (Reducing Asthma Attacks in Children using Exhaled Nitric Oxide) trial. They told us that taking part in RAACENO was a positive experience and that having treatment guided by a breath test and computer was broadly acceptable.

Efficacy and Mechanism Evaluation

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

EME programme

The Efficacy and Mechanism Evaluation (EME) programme funds ambitious studies evaluating interventions that have the potential to make a step-change in the promotion of health, treatment of disease and improvement of rehabilitation or long-term care. Within these studies, EME supports research to improve the understanding of the mechanisms of both diseases and treatments.

The programme supports translational research into a wide range of new or repurposed interventions. These may include diagnostic or prognostic tests and decision-making tools, therapeutics or psychological treatments, medical devices, and public health initiatives delivered in the NHS.

The EME programme supports clinical trials and studies with other robust designs, which test the efficacy of interventions, and which may use clinical or well-validated surrogate outcomes. It only supports studies in man and where there is adequate proof of concept. The programme encourages hypothesis-driven mechanistic studies, integrated within the efficacy study, that explore the mechanisms of action of the intervention or the disease, the cause of differing responses, or improve the understanding of adverse effects. It funds similar mechanistic studies linked to studies funded by any NIHR programme.

The EME programme is funded by the Medical Research Council (MRC) and the National Institute for Health and Care Research (NIHR), with contributions from the Chief Scientist Office (CSO) in Scotland and National Institute for Social Care and Health Research (NISCHR) in Wales and the Health and Social Care Research and Development (HSC R&D), Public Health Agency in Northern Ireland.

This report

The research reported in this issue of the journal was funded by the EME programme as project number 15/18/14. The contractual start date was in February 2017. The final report began editorial review in April 2021 and was accepted for publication in September 2021. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the NHS, the NIHR, the NHR, the NHR, the HTA programme or the Department of Health and Social Care.

Copyright © 2022 Turner *et al.* This work was produced by Turner *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Adviser, Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk